Author's response to reviews

Title: A cross sectional survey of determinants of serum concentrations of polybrominated flame retardants among healthy pregnant women in an urban environment

Authors:

Megan K Horton (mh2180@columbia.edu)
Sabine Bousleiman (sb1080@columbia.edu)
Richard Jones (rjl9@cdc.gov)
Andreas Sjodin (zrq4@cdc.gov)
Xinhua Liu (xl26@columbia.edu)
Robin Whyatt (rmw5@columbia.edu)
Ronald Wapner (rw2191@mail.cumc.columbia.edu)
Pam Factor-Litvak (prf1@columbia.edu)

Version: 3 Date: 15 October 2012

Author's response to reviews: see over
October 11, 2012

Professor Philippe Grandjean, Editor-in-Chief
Department of Environmental Health
401 Park Drive
Landmark Center Room 3-110 East
Boston, Massachusetts 02215

Dear Dr. Grandjean and the EH Editorial Team,

Thank you for considering our submission for publication in your journal. We would like to thank the editors and reviewers for their thoughtful and constructive comments on the manuscript originally entitled “Predictors of serum concentrations of PBDEs among healthy pregnant women in an urban environment”. We have addressed the reviewers’ comments below (in blue) and made changes in the text as appropriate.

We feel these reviews have greatly strengthened the quality and contribution of this manuscript and appreciate your additional review.

We look forward to hearing from you.

Best,

Megan K. Horton, PhD
Department of Epidemiology
Mailman School of Public Health, Columbia University
722 W 168th St, Room 735
New York, New York 10032
Phone: 212-305-8303
Email: mh2180@columbia.edu